Author's response to reviews

Title: Upper limb neuropathy in computer operators? A clinical case study of 21 patients

Authors:

Jorgen R Jepsen (jrj@ribeamt.dk)

Version: 4 Date: 2 August 2004

Author's response to reviews:

1. In the abstract I have changed the title Conclusion to Conclusions.
2. I have changed Reference numbers in the text from superscript to normal script in square brackets.
3. I have stated Authors' contribution as suggested.
4. I have not included an Acknowledgements section
5. I have modified the reference list so that author initials follow surnames.
6. Full stops has been inserted in the figure titles as suggested
7. All figures have been cropped